Author's response to reviews

Title: AIDS knowledge and attitudes in a Turkish population: an epidemiological study

Authors:

Unal Ayranci (unalayranci@yahoo.com)

Version: 5 Date: 25 August 2005

Author's response to reviews: see over
Dear Editorial Office,
I did the following changes as you indicated. Those in red are those being done by me.

First, we do need you to acknowledge your copyeditor.
I can not understand who my copyeditor is? And also how to acknowledge. Could you please explain this?

Second, we require a competing interest statement.
Before “acknowledgements” and after “Conclusion” I added the following sentence:
“Competing interests
The author declare that he has no competing interests”

Third, we require that any funding you received for this study is acknowledged.
I did not receive any funding for this study

Fourth, we require that you add to your section on ethical approval what body approved the work.
The legal ethical consent was changed as the following:
“Legal ethical consent
Ethical permission for the study was obtained prior to collect data, by contacting and receiving approval from the appropriate management authority, the health directorship of the city involved. Participants were assured of the confidentiality of their responses and provided informed verbal consent.”

We would also request that you go through the manuscript formatting checklist one more time and ensure that your revised manuscript conforms to all of the points. The link to the formatting checklist is provided at the bottom of this e-mail.
I was controlled again

With kind regard
Dr. Unal Ayranci